⁶Flash Drought Indicator Intercomparison in the United States

TRENT W. FORD[®],^a JASON A. OTKIN,^b STEVEN M. QUIRING,^c JOEL LISONBEE,^{d,e} MOLLY WOLOSZYN,^{d,e} JUNMING WANG,^a AND YAFANG ZHONG^b

^a Illinois State Water Survey, Prairie Research Institute, University of Illinois Urbana–Champaign, Champaign, Illinois ^b Cooperative Institute for Meteorological Satellite Studies, Space Science and Engineering Center, University of Wisconsin–Madison, Madison, Wisconsin

^c Department of Geography, The Ohio State University, Columbus, Ohio

^d Cooperative Institute for Research in Environmental Sciences, University of Colorado Boulder, Boulder, Colorado ^e NOAA/National Integrated Drought Information System, Boulder, Colorado

(Manuscript received 10 May 2023, in final form 25 August 2023, accepted 27 September 2023)

ABSTRACT: Increased flash drought awareness in recent years has motivated the development of numerous indicators for monitoring, early warning, and assessment. The flash drought indicators can act as a complementary set of tools by which to inform flash drought response and management. However, the limitations of each indicator much be measured and communicated between research and practitioners to ensure effectiveness. The limitations of any flash drought indicator are better understood and overcome through assessment of indicator sensitivity and consistency; however, such assessment cannot assume any single indicator properly represents the flash drought "truth." To better understand the current state of flash drought monitoring, this study presents an intercomparison of nine, widely used flash drought indicators. The indicators represent perspectives and processes that are known to drive flash drought, including evapotranspiration and evaporative demand, precipitation, and soil moisture. We find no single flash drought indicator consistently outperforms all others across the contiguous United States. We do find the evaporative demand- and evapotranspiration-driven indicators tend to lead precipitation- and soil moisture-based indicators in flash drought onset, but also tend to produce more flash drought events collectively. Overall, the regional and definition-specific variability in results supports the argument for a multi-indicator approach for flash drought monitoring, as advocated by recent studies. Furthermore, flash drought research-especially evaluation of historical and potential future changes in flash drought characteristics-should test multiple indicators, datasets, and methods for representing flash drought, and ideally employ a multi-indicator analysis framework over use of a single indicator from which to infer all flash drought information.

SIGNIFICANCE STATEMENT: Rapid onset or "flash" drought has been an increasing concern globally, with quickly intensifying impacts to agriculture, ecosystems, and water resources. Many tools and indicators have been developed to monitor and provide early warning for flash drought, ideally resulting in more time for effective mitigation and reduced impacts. However, there remains no widely accepted single method for defining, monitoring, and measuring flash drought, which means most indicators that are developed are compared with other individual indicators or conditions and impacts in one or two flash drought events. In this study, we measure the state of flash drought monitoring through an intercomparison of nine, widely used flash drought indicators that represent different aspects of flash drought. We find that no single flash drought indicators, datasets, and methods. Furthermore, we suggest flash drought research—especially that which reflects on historical or projected changes in flash drought characteristics—should seek multiple indicators, datasets, and methods for analyses, thereby reducing the potentially confounding effects of sensitivity to a single indicator.

KEYWORDS: Drought; Precipitation; Evapotranspiration; Soil moisture

1. Introduction

Flash drought is characterized by unusually rapid drought intensification over subseasonal time scales, which reduces lead time for preparation, response, and management to drought conditions (Otkin et al. 2022). Although precipitation deficits are a prerequisite for all types of drought, flash drought is often caused by concurrent precipitation deficit and elevated evapotranspiration due to high temperatures, low humidity, strong winds, abundant solar radiation, or a combination of these conditions (Otkin et al. 2018). The co-occurrence of limited precipitation and high evapotranspiration can cause a rapid depletion of soil moisture availability for plant use (Mozny et al. 2012; Otkin et al. 2013; Ford and Labosier 2017; Koster et al. 2019). Rapid soil drying can lead to vegetation moisture stress and the emergence of ecological and agricultural impacts. Because it results from insufficient precipitation and high evapotranspiration, flash drought is most likely to develop during the growing season

DOI: 10.1175/JAMC-D-23-0081.1

© 2023 American Meteorological Society. This published article is licensed under the terms of the default AMS reuse license. For information regarding reuse of this content and general copyright information, consult the AMS Copyright Policy (www.ametsoc.org/PUBSReuseLicenses).

^o Denotes content that is immediately available upon publication as open access.

Supplemental information related to this paper is available at the Journals Online website: https://doi.org/10.1175/JAMC-D-23-0081.s1.

Corresponding author: Trent Ford, twford@illinois.edu

1714

when both evaporative demand and vegetation moisture requirements are high. Consequently, flash droughts have the potential to cause significant ecological, economic, and cultural impacts. For example, economic impacts from the 2012 flash drought in the central United States were estimated to be approximately \$40 billion (Rippey 2015). The 2017 flash drought in the U.S. northern plains caused an estimated \$240 million loss in recreation tourism in Montana alone (Hoell et al. 2020). These are just two of many examples of the adverse impacts of flash droughts on the environment and economy (e.g., Nguyen et al. 2019; Basara et al. 2019; Christian et al. 2020; Hunt et al. 2021).

The substantial impacts caused by recent flash drought events have motivated development of multiple indicators and indices for flash drought monitoring, identification, and early warning (Lisonbee et al. 2021). Mirroring the varied and complex processes through which flash drought occurs (e.g., Osman et al. 2022), indicators have been derived from a diverse set of meteorological and hydrological variables known to contribute to flash drought, including precipitation deficits (Noguera et al. 2020; Fu and Wang 2022), elevated evaporative demand or evapotranspiration (Otkin et al. 2013; Hobbins et al. 2016; Christian et al. 2019), and soil moisture depletion (Osman et al. 2021; Otkin et al. 2021). While individual indicators have been shown to effectively capture the onset and evolution of flash drought events (Otkin et al. 2016; Nguyen et al. 2019), they inherently represent only a single component of the interconnected processes that cause flash drought. The singular focus on precipitation, evapotranspiration, evaporative demand, or soil moisture, limits the extrapolation of these indicators beyond their test cases for monitoring compound multivariate hazards such as flash drought (e.g., Otkin et al. 2013).

The myriad limitations and caveats in any individual flash drought indicator can be better understood and overcome through a combination of case study and climatological assessment of their ability to capture flash drought onset and/or severity. Numerous prior studies have done this, evaluating the flash drought monitoring efficacy of individual indicators across a single or multiple flash drought events (e.g., Anderson et al. 2013; Ford et al. 2015; Christian et al. 2019; Chen et al. 2019; Hoffmann et al. 2021). The advantage of a case study assessment is that it tests an indicator in a real drought situation. However, differences in characteristics between flash droughts also necessitate a climatological assessment to ensure an indicator can perform well across a multitude of conditions and regions. In this study, we leverage the benefits of both climatological and case study assessments to test indicators.

The ideal validation of a drought indicator would be through its comparison with a tangible drought impact (e.g., crop yield loss, watering restrictions, excess wildlife mortality). However, drought impact assessment is a challenging endeavor partly because of the diversity of the impacts and when they occur. For example, the 2012 U.S. flash drought caused widespread mortality of seedlings on Christmas tree farms; however, this impact was not fully realized until several years later with a shortage of Christmas tree stock (Gutzmer 2018). The complexity of drought impacts has challenged impact monitoring and reporting, and our current infrastructure-while rapidly improving-is not sufficient to capture the breadth and depth of impacts from a severe drought (Bachmair et al. 2016). Consequently, most prior studies that evaluate one or more indicators of drought or flash drought determine a benchmark dataset or indicator that is assumed to represent the "truth." This approach may be reasonable in cases where the ultimate goal is improving monitoring or early warning for a specific result, such as soil moisture drought (e.g., Hoffmann et al. 2021; Parker et al. 2021). However, the results of such 1-to-1 indicator comparisons are limited by the uncertainties and limitations of the indicator that serves as the benchmark or "truth." This issue is exacerbated in the case of flash drought, where there is not a single dataset or indicator that is widely considered the benchmark to which other indicators can be compared.

From an operational monitoring perspective, the suite of potentially effective flash drought indicators can act as a complementary set of tools by which to inform flash drought response and management. However, the limitations and uncertainties with each tool must be well measured and communicated between research and operations to ensure proper use of any individual indicator (Otkin et al. 2022). Furthermore, we argue these limitations cannot be properly understood through 1-to-1 indicator comparisons.

Instead of assuming a single indicator or dataset represents the truth of flash drought, for our intercomparison we assume all indicators represent a version of the truth. In this way, we can evaluate a single indicator with respect to the multiple versions of the truth provided by the other indicators. Ideally, an indicator's version of the truth will closely match that of other indicators, providing confidence in a convergence of evidence toward operational flash drought monitoring. Osman et al. (2021) used a similar approach to study the sensitivity of flash drought climatology and trends in individual indicators and definitions. Our study is focused on providing a holistic set of evaluations by which to inform operational flash drought monitoring using a suite of indicators. In this way, the study highlights differences in indicator performance when compared with other indicators and by geographic region. The study's focus is the contiguous United States (CONUS) and evaluates flash drought monitoring over the period from 2002 to 2021, with flash drought case studies in 2012 and 2019.

2. Flash drought indicators

Since its inception in the scientific lexicon in the early 2000s, flash drought has been defined, studied, and monitored using many indicators and datasets. Most flash drought studies have used one or more indicators that can be broadly categorized as those representing 1) evapotranspiration, 2) soil moisture, and 3) precipitation (Lisonbee et al. 2021). Lisonbee et al. (2021) also include a category for temperature-based monitoring, but the effects of temperature on flash drought are captured by the evaporative demand-based indicators. For this study, we select indicators from these three categories, thereby representing many important processes and

TABLE 1. Flash drought indicators used in the study.

Indicator	Native resolution	Period of record	Reference	Data source
EDDI (14 and 28 day)	0.125°	1979-present	Hobbins et al. (2016)	https://psl.noaa.gov/eddi/
ESI (14 and 28 day)	0.125°	2002-present	Anderson et al. (2007)	https://climateserv.servirglobal.net/map
SMVI	0.125°	1979–present	Osman et al. (2021)	https://ldas.gsfc.nasa.gov/nldas/ NLDAS2forcing.php
SPEI (14 and 30 day)	4 km	1979-present	Vicente-Serrano et al. (2010)	https://www.climatologylab.org/gridmet.html
SPI (14 and 30 day)	4 km	1979-present	McKee et al. (1993)	https://www.climatologylab.org/gridmet.html
USDM	_	1979–present	Svoboda et al. (2002)	https://droughtmonitor.unl.edu/DmData/ GISData.aspx

impacts associated with flash drought. Within each category we select indicators that are both widely used in previous research and available for operational monitoring. The latter of these conditions is important as our intent is to remark on the current state of flash drought indicators for real-time monitoring applications. Table 1 contains more detailed information about each of the flash drought indicators used. Although the indicators selected for this study are not meant to be exhaustive, they do include at least one indicator from each of the major flash drought monitoring categories delineated by Lisonbee et al. (2021).

a. Evapotranspiration and evaporative demand indicators

While there has been considerable disagreement among researchers as to how flash drought is characterized and defined, most agree evapotranspiration and evaporative demand play important roles in rapid onset drought and intensification (Otkin et al. 2018; Pendergrass et al. 2020; Nguyen et al. 2023; O and Park 2023). We use two widely studied evapotranspiration and evaporative demand-based indicators for this study: the evaporative demand drought index (EDDI; Hobbins et al. 2016) and the evaporative stress index (ESI; Anderson et al. 2007).

EDDI is a nonparametric percentile-based representation of evaporative demand anomalies. In this study, as in most applications of EDDI, evaporative demand is estimated using the American Society of Civil Engineers standardized reference evapotranspiration equation (Allen et al. 2005), based on the parameterized Penman-Monteith equation. Standardized evapotranspiration probabilities are derived using an inversenormal approximation. Negative (positive) EDDI values indicate evaporative demand that is below (above) the climatological median values. A more detailed description of EDDI, and its derivation are provided in Hobbins et al. (2016). For this study, we obtained daily EDDI datasets from the NOAA/Physical Sciences Laboratory, where EDDI is also available for operational monitoring (https://psl.noaa. gov/eddi/). Reference evapotranspiration used to compute EDDI comes from the operational North American Land Data Assimilation System (NLDAS-2; Xia et al. 2012). Daily EDDI data are available at 0.125° spatial resolution across CONUS from 1979 to present. We use EDDI computed over 14-day (EDDI-14) and 28-day (EDDI-28) intervals to capture

the typical duration of flash drought onset (e.g., Otkin et al. 2018). We also computed all results with 56-day EDDI, but the results did not significantly differ from the 28-day iteration, so we only show EDDI-28 results for brevity.

ESI is an evapotranspiration -based indicator, but unlike EDDI that uses only reference evapotranspiration, ESI is computed as the ratio of the actual evapotranspiration to a reference evapotranspiration. The Atmospheric Land Exchange Inverse (ALEXI; Anderson et al. 1997) model is used to estimate actual and reference evapotranspiration. Ratios of actual to reference evapotranspiration are standardized using a similar procedure as EDDI, based on climatological anomalies. A more detailed description of ESI and its derivation are in Anderson et al. (2007) and Otkin et al. (2013). ESI is available for operational monitoring at ClimateSERV (https:// climateserv.servirglobal.net/map), but only the 28- and 84-day ESI products are available for download. For consistency with EDDI, we obtained 14-day (ESI-14) and 28-day (ESI-28) products directly from Dr. C. Hain at the NASA Marshall Space Flight Center. Daily ESI-14 and ESI-28 products are available at 0.125° resolution across CONUS from 2002 to present.

To identify flash droughts based on ESI and EDDI, we combine the methods of Pendergrass et al. (2020) and Parker et al. (2021). Specifically, a flash drought is identified with a \geq 50-percentile change in EDDI or ESI (toward drying) over 14 days. This onset period must end with EDDI reaching at least the 80th percentile or ESI dipping to or below the 20th percentile. Last, the high EDDI or low ESI values must be sustained for another 14 days to imply drought conditions. A flash drought event was terminated when the EDDI fell below the 80th percentile (or ESI climbed above the 20th percentile) and remained there for at least another 14 days. This technique ensured the method did not identify two separate flash droughts within a 2-week period.

b. Soil moisture indicator

Soil moisture is an important indicator of flash drought (Ford and Labosier 2017; Otkin et al. 2019) and is often used as a benchmark by which to compare other indicators (Christian et al. 2021; Hoffmann et al. 2021; Parker et al. 2021). For this study, we use 0–40-cm soil moisture from the NLDAS-2 Noah model to represent soil moisture conditions. The Noah model simulates hourly soil moisture at multiple depths, and we integrate conditions across 0–40 cm, matching the depths used in past studies (e.g., Ford and Labosier 2017; Osman et al. 2021). Daily NLDAS-2 soil moisture is available at a 0.125° resolution from 1979 to present, and we aggregate hourly soil moisture to daily averages. The NLDAS-2 Noah model was used in this study because it has been previously shown to have high soil moisture simulation fidelity (Xia et al. 2015), and it is available in near real-time for operational drought monitoring (https://ldas.gsfc. nasa.gov/nldas/drought-monitor).

To identify flash drought based on NLDAS-2 soil moisture, we adopt the soil moisture volatility index (SMVI; Osman et al. 2021). Based on the SMVI, a flash drought is identified if 1) the 5-day running average soil moisture falls below the 20-day running average soil moisture for at least 20 days, and 2) by the end of the 20-day period the soil moisture has dropped below its climatological 20th percentile for that time of the year. Soil moisture percentiles were calculated using a 30-day moving window over calendar years, thereby removing the effects of soil moisture seasonality (e.g., Ford and Quiring 2019). Osman et al. (2021) found the SMVI was more stable than previously proposed soil moisture-based metrics that relied solely on the "drop" in soil moisture from the 40th to the 20th percentile. The flash drought was considered to be terminated once the soil moisture climbed above the 20th percentile and remained there for at least 14 days.

c. Precipitation indicator

As with any drought, precipitation deficits play an important role in determining the speed, intensity, and duration of a flash drought event. In this study, we use the standardized precipitation index (SPI; McKee et al. 1993) as a precipitationbased indicator of flash drought. Precipitation fields for SPI were taken from the gridMet dataset (Abatzoglou 2013), which is available daily at an approximately 4-km spatial resolution across CONUS from 1979 to the present. The precipitation fields were upscaled to the common 0.125° spatial resolution of the other indicators. SPI was then calculated as the gammastandardized *n*-day precipitation accumulation. We derive SPI over 14-day (SPI-14) and 30-day (SPI-30) periods to capture the typical flash drought period of onset.

d. Combined indicator

We use the standardized precipitation evapotranspiration index (SPEI; Vicente-Serrano et al. 2010) as an indicator of the combined effects of precipitation and evapotranspiration. Precipitation and reference evapotranspiration fields for SPEI were taken from the gridMet dataset, and upscaled to the common 0.125° spatial resolution of the other indicators. SPEI was calculated as the log-logistic standardized *n*-day accumulation of differences between precipitation and reference evapotranspiration. We derived SPEI over 14-day (SPEI-14) and 30-day (SPEI-30) periods.

We included both SPI and SPEI in this study to contrast the performance of similar metrics with and without the inclusion of reference evapotranspiration. This comparison is particularly important given the outsized role evaporative demand can plan in flash drought onset (Christian et al. 2020; Noguera et al. 2022). We adopt the method of Noguera et al. (2020) for identifying flash drought based on SPI and SPEI, where a flash drought requires a decrease in SPI or SPEI over 28 days with a total change in SPI or SPEI of 2 or greater and a final SPI or SPEI value of -1.28 or less.

e. The U.S. Drought Monitor

The U.S. Drought Monitor (USDM; Svoboda et al. 2002) is a weekly map-based drought severity analysis produced by one of multiple authors with significant local input on drought conditions. The USDM uses a "convergence of evidence" approach to drought monitoring, such that USDM-depicted drought conditions in any given region on any given week will generally not be based on a single indicator. While somewhat subjective, this approach to drought monitoring reduces the negative impacts of biases in individual indicators and ensures local experts can inform the authors about conditions in their areas. USDM maps are produced each week on Tuesday and are released on Thursday.

We obtained weekly USDM maps in shapefile format and converted them to raster format with the common 0.125° spatial resolution of the other indicators. The shapefile-to-raster function assigned each grid cell the worst USDM drought category that encompassed at least 25% of the gridcell area. The once-weekly USDM dataset was then converted to a daily temporal resolution by assuming the same USDM value for each of the 6 days following the Tuesday on which the USDM map was produced. In this way we compare the USDM as it would be compared in an operational drought monitoring environment, assuming the map depicts drought accurately until the next USDM map is released. Flash drought is then identified in the daily USDM dataset using the Pendergrass et al. (2020) method of having a 2+-category decline in 2 weeks that is then sustained another 2 weeks.

f. Alternative EDDI indicator

Methods used to identify flash drought for both operational monitoring and research applications generally follow the same guidelines irrespective of the indicator or variable used. Those guidelines tend to include a "flash" component where conditions change from nondrought to drought in a relatively short period of time, typically 2–6 weeks. Also included is a "drought" component where conditions must decline to at least drought status, and in many cases persist in drought status for another 2–6 weeks. However, differences in variables, thresholds, time periods, and standardization techniques can cause significant differences in the climatological frequency and intensity of flash drought across different regions. To better understand the sensitivity of our results to the method of identifying flash drought, we repeat the analysis using an alternative identification method based on EDDI-14.

Following a standardized change anomaly technique from Otkin et al. (2013) and Christian et al. (2019), we derive a "flash" component by calculating a standardized change in EDDI (EDDI_{*ijd*z), such that}

$$EDDI_{ijd}z = \frac{EDDI_{ijd} - \overline{EDDI}_{ijd}}{\sigma EDDI_{ijd}},$$
(1)

where EDDI_{ijdZ} is the daily change in EDDI-14 at grid cell *i*, *j* and calendar day *d*, $\overline{\text{EDDI}_{ijd}}$ is the mean change in EDDI-14 for a specific grid cell and calendar day, and σEDDI_{ijd} is the standard deviation of EDDI-14 for a specific grid cell and calendar day. Therefore, EDDI_{ijdZ} represents the *z* score of the change in EDDI-14 for a specific grid cell and calendar day. This method accounts for the mean and variability of day-to-day change in EDDI-14 in a specific location, thereby removing the background climatology of high frequency changes in EDDI-14.

Flash drought is identified based on $\overline{\text{EDDI}_{iid}}z$ (herein referred to as $\Delta EDDI_z$) using the 4-part method proposed by Christian et al. (2019). Flash drought events are required to have 1) a minimum length of change of 30 days and 2) a final EDDI-14 value at or above its 80th percentile at the end of the 30-day "flash" period. Additionally, the Δ EDDI_z must be at or above the 60th percentile between individual days and there must be no more than 6 days in the 30-day period where the daily ΔEDDI_z value is below the 60th percentile. Christian et al. (2019) calculated flash drought at the pentad scale and required that at least 5 of 6 pentads during the "flash" period had a change above the 60th percentile. To match the daily scale of our analysis, we similarly require at least 24 of 30 days in the "flash" period to have $\Delta EDDI_z$ values above the 60th percentile. The $\Delta EDDI_z$ flash drought climatology and intercomparison statistics are compared with those from EDDI-14 to evaluate the sensitivity of our results to flash drought identification methods. The goal of this analysis is not to provide a comprehensive evaluation of all indicator sensitivity to flash drought, but instead demonstrate the importance of understanding threshold sensitivity when using a single indicator for flash drought monitoring and/or research.

Standardization and percentile calculation for EDDI, SPEI, SPI, and SMVI used statistics over the common period 1979– 2021, while the ESI products used were standardized based on statistics from the 2002–21 period. The different time periods over which indicators were computed could introduce a bias ratio in the results; however, given all but 1 of the standardized products used the same common period 1979–2021, it would be unfair to restrict this period because the ESI solely uses a shorter period. All indicators were compared at a common 0.125° spatial scale and daily temporal scale over the common period of March–November 2002–21. Because flash drought is predominantly a warm-season phenomenon, we did not assess flash drought onset between December and February.

3. Intercomparison methods

a. Climatological assessment

The goal of this study is to compare ten flash drought indicators to better understand the consistency between them when monitoring for flash drought. We hold a single indicator as the "comparison," which is the indicator we are evaluating. This indicator is then compared with each of the other nine indicators separately. In these comparisons, we assume each of the nine other indicators is the "benchmark" or truth. For each "comparison" and "benchmark" indicator evaluation, the 1-to-1 and aggregated comparison statistics are reported. This process is then repeated for each of the nine remaining indicators, iteratively holding one as the "comparison" and the others as the "benchmark." This process allows us to evaluate each indicator against other individual indicators as well as against the field of flash drought indicators in aggregate.

One-to-one indicator evaluations are done using a map comparison approach. This process is described using the following example comparing EDDI-14 to SMVI. In this case, EDDI-14 is selected as the "comparison" indicator (the indicator we want to evaluate) and the SMVI is selected as the "benchmark" indicator. For those more familiar with forecast verification, in this example the SMVI is practically denoted as the observation and EDDI-14 as the prediction. We then select a single day's field of EDDI-14 across CONUS and compare it with each single day's field of SMVI within 4 weeks before and after the EDDI-14 day selected. In this way, the EDDI-14 field is compared with every SMVI field from 28 days prior to 28 days after. Our 1-to-1 indicator comparisons lag and lead the benchmark indicator to capture the potential differences in the timing of flash drought onset between indicators. This approach more fairly evaluates indicators that may capture the same flash drought within a relatively small time window, and also allows us to measure the average lag or lead time with which the comparison indicator captures flash drought.

For each 1-to-1 comparison of daily flash drought fields, each grid cell in the comparison indicator across CONUS is compared with the corresponding grid cell in the benchmark indicator. The comparison produces a contingency table of hits a, false alarms b, misses c, and correct negatives d. A hit is when both indicators indicate a flash drought, and a correct negative is when both indicators indicate that there is not a flash drought. In the case of our analysis, there is no certified flash drought "truth," and therefore there are no true false alarms or misses. Instead, both false alarms and misses represent disagreement between the comparison and benchmark indicators. The contingency table of each 1-to-1 comparison is used to calculate the threat score (TS):

$$TS = \frac{a}{a+c+b}.$$
 (2)

The TS is a measure of how well flash drought events in the comparison indicator, EDDI-14 in our example, correspond with flash drought events in the benchmark indicator, SMVI in our example. We also compute the bias ratio measure for each 1-to-1 comparison:

bias ratio
$$= \frac{a+b}{a+c}$$
. (3)

Bias ratio is a measure of the ratio of flash drought events in the comparison indicator to the frequency of flash drought events identified by the benchmark indicator. We use bias ratio to indicate if the comparison indicator tends to identify fewer or more flash droughts with respect to other indicators. These comparisons and calculations of TS and bias ratio are repeated by comparing the EDDI-14 field on the single day to every SMVI field from 28 days prior to 28 days after. We then report the highest overall TS between EDDI-14 and SMVI across the 57-day comparison and the bias ratio on the day with the highest TS.

This approach has the benefit of simultaneously comparing two indicators over space and over time and producing 1) the overall strongest correspondence between the indicators (i.e., highest TS) and 2) the time at which the highest correspondence occurred. The timing of the strongest correspondence gives us important information on the lead or lag of the corresponding indicator relative to the benchmark indicator. For example, if the strongest correspondence between EDDI-14 and SMVI occurs when comparing EDDI-14 on day n to SMVI on day n + 3, the implication is that EDDI-14 depiction of flash drought led SMVI flash drought by 3 days. The maximum TS, the bias ratio on the day of the maximum TS, and the timing of the maximum TS are recorded. The entire process is then repeated for each daily field between March 2002 and November 2021 excluding winter months. Results are reported as the overall average of daily maximum TS values and their corresponding bias ratio values, and the distribution of the timing of maximum TS between comparison and benchmark indicators. These aggregate statistics provide an overall summary of comparison between two indicators.

We use a "neighborhood maximum" approach (e.g., Sobash et al. 2011; Schwartz and Sobash 2017) to relax the requirement of an exact match between comparison and benchmark indicators at the grid cell for a hit. Specifically, a hit is recorded for a grid cell if both the comparison and benchmark indicators show flash drought anywhere in a surrounding 5×5 gridcell neighborhood. Such neighborhood-based approaches have been used widely in high-resolution precipitation forecast verification (e.g., Ebert 2008) to acknowledge the unrealistic expectation of a perfect 1-to-1 forecast at high resolution. While flash drought monitoring, and drought monitoring more broadly, is typically over coarser spatial scales, the same issues can arise from 1-to-1 gridcell comparisons that expect perfect matches between indicators. In most cases of flash drought monitoring, the precision of flash drought identification is on the order of tens of kilometers. Therefore, we selected a 5 \times 5 gridcell neighborhood for our evaluation, with an approximate spatial footprint of 60 km \times 60 km, which was deemed small enough to provide useable information for flash drought monitoring but coarse enough to provide a fair comparison between indicators. The neighborhood maximum approach was found to be the most consistent among 3 neighborhood-based comparison approaches for precipitation forecasting (Schwartz and Sobash 2017) and was deemed a good fit for our purpose.

One important detail to note is that we only calculated TS and bias ratio between two indicators when flash drought was indicated on at least 1% of the CONUS or CONUS region area in the comparison indicator. This was done to eliminate erroneously high TS values arising from shared flash drought between the two indicators in one or two grid cells, with no flash drought throughout the vast majority of CONUS. The comparisons were completed for all combinations of indicators across CONUS, and then repeated on a regional basis. Comparisons were also completed across each of the seven National Climate Assessment regions (defined in Fig. S1 in the online supplemental material), allowing evaluation of regional differences in performance.

b. Case studies

We use two noteworthy flash drought events as case studies by which to compare the indicators. The first is the 2012 central U.S. flash drought, one of the most widely studied flash drought events globally (Otkin et al. 2016; Basara et al. 2019; Kam et al. 2021). The 2012 drought began in late spring following an extremely warm March and April, and, encompassed much of the central United States by July. Agricultural impacts from the 2012 event are estimated to have exceeded \$30 billion (Rippey 2015) and caused a shock to global food supply (Boyer et al. 2013). We examine the change in drought conditions represented in each drought indicator from 1 March to 1 September 2012 across a portion of the central United States that was significantly impacted by the drought event. The second flash drought we use as a comparison case study is the 2019 event in the southeastern United States. While not as infamous as the 2012 central U.S. event, the 2019 southeastern flash drought developed quickly and had significant impacts on agriculture and water resources (Di Liberto 2019). We examine the change in drought conditions represented in each drought indicator from 1 August to 1 November 2019 across a portion of the southeastern United States that was significantly impacted by the drought event.

The case study analyses include both map-based intercomparison across the lifetime of each flash drought and a time series analysis of flash drought spatial extent across the respective regions affected by each event.

4. Results

a. Flash drought climatological frequency

The methods for identifying flash drought in all 10 indicators are based on a change in percentile or standardized anomaly over a 2-4-week period, persisting for another 2-4 weeks. Despite the similarity in flash drought identification methods, there are large differences in flash drought frequency between the 10 indicators (Fig. 1). In general, the indicators aggregated over 14-day periods had a higher frequency than the same indicators aggregated over 28- or 30-day periods. This is likely because 14-day aggregated indicators can respond more quickly to evolving environmental conditions, which produces overall more flash drought events. Specifically, the SPEI-14 shows the overall highest frequency of flash drought across CONUS over the 20-yr study period, with a domain average of 39 events, followed by EDDI-14 with 34 events on average, SPEI-30 with 29 events, and SPI-14 with 28 events. All indicators except the USDM (17 events on average) identified an average of at least 1 flash drought event per year per grid cell across CONUS. The EDDI-14 indicated a higher overall flash drought frequency than ESI-14, which is likely due to EDDI's use of reference evapotranspiration in contrast to ESI's actual evapotranspiration. Short-lived spikes in evaporative demand could induce a flash drought indication based on EDDI-14. But if soil moisture is sufficient to meet that additional demand, there may



FIG. 1. Maps of the climatological frequencies of flash drought events between 2002 and 2021. The numbers shown are the flash drought event frequencies for each indicator.

not be an equivalent response in actual evapotranspiration, thereby producing a differential response between EDDI and ESI. These differences seem to be less apparent in the 28-day aggregation of EDDI and ESI, as their CONUS-wide climatological flash drought frequency is similar.

The maps of climatological flash drought frequency also show considerable regional variability for many indicators. EDDI-14 and EDDI-28 both exhibit a relatively low flash drought frequency over the southeastern United States, but in contrast, ESI-14, ESI-28, and SMVI all have relatively high frequency in the same region (Fig. 1). Both SPI-14 and SPI-30 identified fewer than 10 flash drought events in the southwest United States, including most of California and Nevada; however, their SPEI counterparts show relatively high flash drought frequency in that region. These differences are likely due to the SPEI's inclusion of potential evapotranspiration, which is a particularly important component to moisture balance and flash drought occurrence in dry regions. The USDM shows the highest frequency of flash drought in the Southern Plains and Southeast regions, with relatively few events identified in the Northwest, Southwest, and Northeast.

b. Indicator intercomparison across CONUS

The daily maximum TS between two indicators is then averaged across all days of the comparison between 2002 and 2021. The overall TS results of all 1-to-1 indicator comparisons are shown in Fig. 2. Higher TS values indicate better correspondence, a TS value of 0 indicates no agreement and a TS value of 1 represents a perfect match. Overall, the various indicators are moderately well related to one-another, with TS



FIG. 2. Color table of the average TS from comparisons between two indicators across all comparisons. The far-right column shows the median of the nine TS values for each indicator.

values between 0.2 and 0.6. As expected, the different integration periods of the same indicator (e.g., EDDI-14 and EDDI-28) show good agreement, as are SPI and SPEI integrated over the same period (14 or 30 days). However, there is also a notably strong correspondence between EDDI-14 and SPEI-14 (0.33). Some of the indicators with relatively weak correspondence included EDDI-28 and SPEI-30 (0.19) and ESI-28 and SPI-30 (0.20). All indicators had a noticeably lower correspondence with the USDM, with TS scores ranging from 0.11 to 0.13. The far-right column of the table in Fig. 2 shows the median of the nine intercomparisons for each indicator. The SPEI-14 had the highest overall TS (0.29), followed by SPI-14 (0.27) and EDDI-14 (0.26). The EDDI, ESI, SPEI, and SPI indicators derived over 2 weeks consistently performed better than those derived over 4 weeks or 30 days. The only indicator based on soil moisture-the SMVI-had a higher correspondence to ESI-14 and SPEI-14 (0.28 and 0.29, respectively) compared with ESI-28 and SPEI-30 (0.21 and 0.22, respectively).

For each comparison, we report the bias ratio at the lead-lag comparison when the TS is maximized between indicators. For example, if the TS between SPEI-14 and SPI-14 is maximized

at a 1-day lead time, we will report the bias ratio between the two indicators at that lead time. The bias ratio signals whether the comparison indicator tends to identify more or fewer flash drought events relative to the benchmark indicator. The bias ratio metric is then averaged over all daily comparisons between two indicators, with the results shown in Fig. 3. Bias ratio values over 1 (under 1) signify the comparison indicator indicates more (fewer) flash droughts relative to the benchmark indicator. The last column in Fig. 3 shows the median of the nine bias ratio values for each indicator. Overall, the indicators integrated over 14-day periods tend to have a higher flash drought frequency relative to other indicators. The ESI-14 has a bias ratio much closer to 1 (1.03). On the other hand, ESI-28 exhibited the lowest median bias ratio (0.69). SMVI and EDDI-28 also had lower bias ratio values, 0.75 and 0.76, respectively, which represents fewer flash drought events overall from those indicators (Fig. 3).

For each 1-to-1 indicator match, the map of the comparison indicator on day n was compared with the maps of the benchmark indicator are days n - 28 to n + 28, and the day of the benchmark indicator that had the highest TS with the comparison



FIG. 3. Color table of the average bias ratio from comparisons between two indicators across all comparisons. The far-right column shows the median of the nine bias ratio values for each indicator.



FIG. 4. Violin plots of the distribution of the timing of maximum correspondence between each indicator and all other indicators. Each point in the distribution represents one daily comparison. The white points in the plots represent the overall average timing, and the dark vertical line shows the interquartile range. Negative timing values represent a leading indicator; positive values represent a lagging indicator.

indicator was noted. The distributions of the timing of highest TS between indicators imply whether a given indicator tends to lead or lag other indicators' depiction of flash drought. Figure 4 shows violin plots of the distributions of the timing of highest TS by indicator. The shaded area in each violin plot represents the shape of the distribution of the lead or lag times at which hits occur. The width of the violin plot represents the density of the distribution for those values, such that a wider (narrower) shape indicates more (fewer) hits at that lead or lag time. The dark, vertical line in each plot represents the interquartile range and the white dot represents the distribution median. EDDI-14 and ESI-14 tend to be leading indicators meaning they had the strongest correspondence with other indicators prior to the other indicators showing flash drought. Specifically, the EDDI-14 and ESI-14 lead other indicators by, on average, 3.6 and 2.9 days, respectively. In contrast, the EDDI-28, ESI-28, and SMVI lagged other indicators by, on average, 4.1, 3.9, and 3.7 days, respectively. SPEI-14 tended to lag other indicators by less than 1 day, while the other three SPEI or SPI metrics tended to be leading indicators as often as they were lagging indicators. The USDM had the broadest distribution of timing, but on average tended to lag by 2.5 days.

Overall, the intercomparison of flash drought indicators across CONUS elucidated several important findings. Evapotranspiration and evaporative demand-based indicators integrated over 2-week periods (EDDI-14 and ESI-14) tend to lead other indicators in their identification of flash drought. However, while the EDDI-14 also produced many more flash droughts than most other indicators, ESI-14 tended to show approximately the same flash drought frequency as other indicators (i.e., bias ratio close to 1), and ESI-14 had a similar overall TS value as EDDI-14 (0.25 vs 0.26). SPEI-14 had the overall highest TS values, but also tended to identify many flash droughts that were not identified by other indicators and did not consistently lead other indicators in flash drought, as did EDDI-14 and ESI-14. The SMVI, based entirely on soil moisture, was consistently a lagging indicator, as is well documented in process-based flash drought studies (Otkin et al. 2016; Ford and Labosier 2017). Across CONUS, the EDDI-14, ESI-14, and all SPI and SPEI indicators tended to produce many more flash droughts than were identified by SMVI. For example, both EDDI-14 and SPEI-14 identified approximately 1.75 flash droughts for every 1 flash drought identified by SMVI. This result reflects the lack of definite translation of meteorological drought—driven by precipitation deficits and enhanced evaporative demand—to agricultural drought across CONUS.

c. Regional indicator intercomparisons

Comparisons between flash drought indicators were repeated for each of the seven National Climate Assessment regions to evaluate regional variability in indicator performance. Figure 5a shows the average TS values by indicator for each of the seven regions. The SPEI-14, SPI-14, and EDDI-14 indicators consistently have the highest TS values across regions, while the USDM and, to a lesser extent, EDDI-28, ESI-28, and SPEI-30, have lower overall regional scores. There are differences in the overall performance of indicators by region. For example, the median TS value in the Southeast region is 0.27 as compared with a median of 0.21 in the Southwest. Practically, these differences represent stronger correspondence between indicators in the Southeast than in the Southwest region.

We also aggregate bias ratio between indicators by region (Fig. 5b). The patterns of flash drought frequency are mostly similar between regions. SPEI-14 identifies more flash droughts overall in every region, but most so in the Northeast and Southwest regions where the indicator shows 1.95 and 2.27 flash drought events for every 1 event in other indicators. EDDI-28, ESI-28, and SMVI tend to indicate fewer flash droughts in every region. All but the Northeast and Southwest regions have a median bias ratio value very close to 1. The individual region tables

(a)				R	egiona	Threa	t Score	9			-
Northeast	0.25	0.21	0.24	0.22	0.24	0.3	0.22	0.28	0.22	0.12	0.23
Southeast	0,32	0.21	0.3	0.26	0.28	0.32	0.23	0.32	0.25	0.12	0.27
Midwest	0.27	0.22	0.26	0.23	0.26	0.29	0.21	0.3	0.24	0.12	0.25
No. Plains	0.28	0.23	0.27	0.24	0.24	0.28	0.23	0.27	0.24	0.12	0.24
So. Plains	0.27	0.2	0.25	0.22	0.23	0,26	0.2	0.25	0.22	0.12	0.23
Northwest	0.3	0.25	0.25	0.22	0.24	0.29	0.25	0.25	0.24	0.12	0.25
Southwest	0.26	0.21	0.25	0.22	0.21	0.26	0.2	0.19	0.2	0.11	0.21
Median	0.27	0.21	0.25	0.22	0.24	0.29	0.22	0.27	0.24	0.12	0.24
EDD	EDD	1-28 ES	1-14 ES	1-28 5	Bia	S-Reg	ional	1-14 SP	1-30 US	SON ME	dian
EDD	EDD	1-28 ES	1-14 ES	-28 5	NNISPE	I-14 SPE	1-30 SP	1-14 SP	1-30 US	SOMME	dian
b) Northeas	1-14 EDD	1.06	1.25	0.96	MVI SPE Bia: 0.75	5PE s - Reg 1.95	1-30 SP ional 1.32	2.01	1.33	3DN N6	1.29
ED ^C b) Northeas Southeas	st 1.98	1.06 0.35	1.25	0.96 0.68	6.75 0.93	1-14 SPE S - Reg 1.95 1.66	ional 1.32 0.96	2.01	1.33 1.15	0.67	1.29 - 1.01 -
b) Northeas Southeas Midwes	N-14 EDD st 1.98 st - 0.91 st - 1.47	1.06 0.35 0.66	1.25 1.06 0.86	0.96 0.68 0.65	6.75 0.93 0.83	1-14 SPE s - Reg 1.95 1.66 1.72	1.30 SF ional 1.32 0.96 0.9	2.01 1.71 1.96	1.33 1.15 1.28	0.67 0.67 1.71 1.2	1.29 - 1.01 - 1.05 -
b) Northeas Southeas Midwes	1-14 EDD st 1.98 st - 0.91 st - 1.47 s - 1.15	1.06 1.06 0.35 0.66 5 0.53	1.25 1.06 0.86 0.71	0.96 0.68 0.65 0.49	MVI SPE Bia: 0.75 0.93 0.83 0.61	1.95 1.66 1.72 1.37	ional 1.32 0.96 0.9 0.97	2.01 1.71 1.96 1.42	1.33 1.15 1.28 1.12	0.67 0.67 1.71 1.2 1.01	1.29 - 1.01 - 1.05 - 0.99 -
b) Northeas Southeas Midwes <u>5</u> No. Plain	1-14 EDD st 1.98 st - 0.91 st - 1.47 s - 1.15 s - 1.19	1.06 1.066 0.666 5.0.53 0.44	1.25 1.06 0.86 0.71 0.95	0.96 0.68 0.65 0.49 0.58	Bias 0.75 0.93 0.83 0.61 0.69	1.95 1.66 1.72 1.37 1.4	x ³⁰ gr ional 1.32 0.96 0.9 0.97 0.8	2.01 1.71 1.96 1.42 1.43	1.33 1.15 1.28 1.12 1.3	0.67 1.71 1.2 1.01 2.21	1.29 1.01 1.05 0.99 1.07
b) Northeas Southeas Midwes No. Plain So. Plain Northwes	1-14 EDD st 1.98 st -0.91 st -1.47 s -1.15 s -1.19 st -1.27	12 ⁸ ES 1.06 0.35 0.66 0.53 0.44 0.64	1.25 1.06 0.86 0.71 0.95 0.78	0.96 0.68 0.65 0.49 0.58 0.53	8ia: 0.75 0.93 0.83 0.61 0.69 0.72	1.95 1.66 1.72 1.37 1.4 1.54	x ³⁰ sF ional 1.32 0.96 0.9 0.97 0.8 1.21	2.01 1.71 1.96 1.42 1.43 1.08	1.33 1.15 1.28 1.12 1.3 0.96	0.67 1.71 1.2 1.01 2.21 0.84	1.29 - 1.01 - 1.05 - 0.99 - 1.07 - 0.9 -
b) Northeas Southeas Midwes <u>5</u> No. Plain Worthwes Southwes	1.14 EDD st 1.98 st -0.91 st -1.47 s -1.15 s -1.15 s -1.19 st -1.27 st -1.27	128 ES 1.06 0.35 0.66 0.53 0.64 0.64 0.85	1.25 1.06 0.86 0.71 0.95 0.78 1.63	0.96 0.68 0.65 0.49 0.58 0.53 1.04	Bia: 0.75 0.93 0.83 0.61 0.69 0.72 0.96	1.95 1.95 1.66 1.72 1.37 1.4 1.54 2.27	x ⁻³⁰ sp ional 1.32 0.96 0.9 0.97 0.8 1.21 1.62	2.01 1.71 1.96 1.42 1.43 1.08 0.77	1.33 1.15 1.28 1.12 1.3 0.96 1.14	0.67 1.71 1.2 1.01 2.21 0.84 1.26	1.29 - 1.01 - 1.05 - 0.99 - 1.07 - 0.9 - 1.2

FIG. 5. Color tables of (a) average TS values and (b) bias ratio values by indicator for each of the seven National Climate Assessment regions.

of TS and bias ratio are available in Figs. S2 and S3 in the online supplemental material.

The overall patterns of timing of flash drought in each indicator are very similar between regions. In all regions, EDDI-14 and ESI-14 are leading indicators, with average lead times of 2 to 6 days. EDDI-28, ESI-28, and SMVI are lagging indicators in all regions, with lagging times of 2 to 8 days. SPI and SPEI indicators tend to lead and lag at roughly the same rate, and the USDM is a lagging indicator in every region except the Northwest, where it has an average lead time of just under 1 day.

The USDM flash drought method is based on categorical change over 2-week periods, suggesting the USDM changes more quickly in the Northwest region—relative to other indicators tested—than in other regions. This is particularly the case when compared with the Northeast region, where the USDM lags by an average of almost 8 days. It is worth mentioning that most flash droughts in the Northwest region in our study period are in Montana, with fewer than 10 flash droughts identified by the USDM in Washington, Oregon, and Idaho. Meanwhile, the USDM identified fewer than 10 flash droughts in most of the Northeast region. Therefore, these differences may also partly be a factor of relatively small sample sizes in these regions.

d. Sensitivity to flash drought definitions

To better understand the sensitivity of our results to the method of flash drought identification, we repeat our analyses using an alternative flash drought method based on 14-day EDDI, the Δ EDDI_z. The climatology and intercomparison results from $\Delta EDDI_7$ are compared with those from EDDI-14 (Fig. 6). The most striking difference between the two indicators, both based on the same 2-week EDDI dataset, is in the climatological frequency of flash drought over the 20-yr study period (Fig. 6a). While EDDI-14 tends to show higher flash drought frequency in the Northeast, western plains, and Southwest and lower flash drought frequency in the Southern Plains and Southeast, the climatological flash drought frequencies based on $\Delta EDDI_z$ are nearly opposite. $\Delta EDDI_z$ shows a hotspot of flash drought frequency in the Southern Plains, Southeast, and Midwest regions, with relatively few flash droughts in the Northeast, Northwest, and parts of the Southwest regions. For example, the average $\Delta EDDI_z$ flash drought frequency in the Southeast region is 45 events (approximately 2.25 events per year per grid cell), as compared with an average EDDI-14 frequency of just 23 events in the same region (approximately 1.15 events per year per grid cell). Meanwhile, EDDI-14 shows an average of 32 events in the Northeast region (1.6 events per year per grid cell) as compared with 5 events on average based on ΔEDDI_{7} (0.25 events per year per grid cell).

We compared ΔEDDI_z with all other indicators using the same methods as our other intercomparisons, calculating TS, bias ratio, and the timing of flash drought hits. Based on TS, the ΔEDDI_z slightly outperforms EDDI-14 in the Northeast, Southeast, Midwest, and Southern Plains regions, while slightly



FIG. 6. (a) Maps of the climatological frequency of flash drought events between 2002 and 2021 identified by (left) EDDI-14 and (right) Δ EDDI_z definitions. (b) Bar plots of the CONUS- and region-average TS and bias ratio of flash drought identification of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI_z and (red) EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of the timing of hits of (blue) Δ EDDI-14. (c) Violin plots of (blue) Δ

underperforming in the northern plains, Northwest, and Southwest (Fig. 6b). Relative to the large bias ratio of EDDI-14 in the Northeast and Southwest regions, $\Delta EDDI_7$ has bias ratio much closer to 1 and has far fewer flash drought events identified in the Northeast region. The overall higher frequency of flash drought in the Southeast and Southern Plains regions result in higher bias ratio in $\Delta EDDI_z$ than in EDDI-14, but overall, across CONUS the $\Delta EDDI_z$ has a bias ratio closer to 1 than the EDDI-14. These results suggest the Δ EDDI_z may more closely match the other nine indicators in the study; however, these results vary considerably by region. Last, the violin plots in Fig. 6c show the distributions of the timing (i.e., lead or lag) of flash drought hits between $\Delta EDDI_{2}$, EDDI-14, and the other nine indicators. The $\Delta EDDI_7$ distribution tends to be broader than that of EDDI-14, with a longer average lead time of 4.1 days as compared with 3.6 for EDDI-14.

e. Case study intercomparison

Our climatological analysis is complemented by an analysis of flash drought indicators based on two case study events, the 2012 central U.S. and 2019 southeastern U.S. flash droughts. For each case study we examine the intensity and spatial extent of drought across a subregion of the United States, and the change in

characteristics during the drought evolution. The 2012 drought was preceded by a very warm early spring in the central United States that corresponded with elevated evaporative demand (Fuchs et al. 2012). EDDI anomalies were exceedingly high as early as 1 March 2012, representing strong evaporative demand (Fig. 7a). SPEI, SPI, and ESI-28 also showed somewhat- to verydry conditions early in spring 2012, while SMVI only showed drier than normal soil conditions in the western half of the study region. EDDI, SPEI, and SPI indicated rapid drying across virtually the entire region between 1 May and 1 June in 2012, while SMVI, ESI, and USDM indicators only showed drought-like conditions in a fraction of the region (Fig. 7a). The SMVI and USDM showed rapid drought onset or intensification between 1 June and 1 July, better aligning with SPI and SPEI indicators; however, ESI did not show drought across most of the region until later in July. A brief reprieve from the very high evaporative demand reduced EDDI-based drought intensity around 1 July, but EDDI rapidly reintensified in August. By 1 September, rain from Hurricane Isaac had improved SPI and SPEI indicators in the southern half of the region, but the other indicators continued to show intense drought across much of the region.

We also include a time series of flash drought spatial extent across the region as an alternative way of visualizing the evolution



FIG. 7. (a) Maps of indicator conditions from 1 Mar to 1 Sep 2012 in the central United States. (b) The time series of flash drought spatial extent from 1 Mar to 1 Sep 2012. Flash drought extent is expressed as a percent of the central U.S. study region.

of flash drought in 2012 (Fig. 7b). The time series shows a rapid increase in flash drought in EDDI-14 in early March, followed by a plateau and reintensification in May and early June. EDDI-14 shows over 30% of the region in flash drought as of 5 March, which is more than a month before another indicator shows the same flash drought extent (SMVI, 11 April). In general, most indicators show flash drought expansion between mid-April and mid-June, with the SMVI, ESI-14, SPI-14, and SPEI-14 generally leading the ESI-28, SPI-30, and SPEI-30. The USDM lagged most of the other indicators by 2 to 4 weeks prior to 1 June, but then shows a rapid flash drought expansion between June and July and shows 100% of the region experienced a flash drought by 12 July. The EDDI-28, SPEI-30, and SPI-30 indicate flash drought in less than 30% of the region through the end of August, which is considerably less extensive than the other indicators. However, these three indicators do show extensive and intense dry conditions across the region as of 1 June and beyond

(Fig. 7a). Therefore, the lack of flash drought extent—based on EDDI-28, SPEI-30, and SPI-30—is more likely due to the method of flash drought identification rather than the indicators themselves. Irrespective of these issues, there is a convergence of evidence from most indicators that the region experienced extensive flash drought between April and July 2012.

The second case studied differs from the first in many ways. First, the region affected—the Southeast—spans humid subtropical and humid continental climates with much less distinct precipitation seasonality than the central United States. Second, the drought occurred in late summer and autumn, spanning harvest of summer crops and planting of autumn and winter crops. August 2019 began with some small areas of very dry soil and low (dry) values of SPI, SPEI, and ESI in the southern parts of the region (Fig. 8a). Above normal temperatures increased evaporative demand in August and September,



FIG. 8. (a) Maps of indicator conditions from 1 Aug to 30 Oct 2019 in the southeastern United States. (b) The time series of flash drought spatial extent from 1 Aug to 30 Oct 2019. Flash Drought extent is expressed as a percent of the southeastern U.S. study region.

as evidenced by positive EDDI-14 and EDDI-28 anomalies. Like the 2012 drought, EDDI indicated rapidly deteriorating conditions weeks ahead of most other indicators. However, SMVI, SPEI-14 and SPI-14 showed rapid expansion of very dry conditions in the first half of September (Fig. 8a). SPI-30, SPEI-30 and ESI and USDM indicators lagged in the expansion of drought conditions by 2–3 weeks, showing similar expansion in late September. Heavy precipitation from Tropical Storms Nestor and Olga in mid- to late October dramatically improved SPI- and SPEI-based conditions and reduced the extent of drought in the USDM. However, much of the region remained very dry through the end of October according to the SMVI and longer-term EDDI-28.

All indicators show a graduate increase in flash drought extent across the region in August 2019 (Fig. 8b). SPEI-14 and SPI-14 show the quickest increase in flash drought extent in

August, followed by a rapid expansion in mid-September. EDDI-14, ESI-14, ESI-28, and SMVI show steady increasing flash drought extent between late August and late September, while SPEI-30, SPI-30, and USDM have very little expansion of flash drought prior to 15 September, followed by rapid expansion between 20 and 30 September. Despite showing extensive, severe dry conditions across most of the region, EDDI-28 identified flash drought in less than 10% of the region through the end of October, whereas USDM showed over 90% of the region experienced a flash drought between August and October. The SPI-14 and SPEI-14 indicators showed flash drought in at least 40% of the southeastern U.S. region approximately 2 weeks ahead of the USDM (Fig. 8b), as compared with a 3-week lag between SPI-14/SPEI-14 and USDM meeting the same 40% threshold in the 2012 drought (Fig. 7b).

1726

5. Discussion

a. Challenges of a single-indicator focus

Most flash drought studies characterize a noteworthy flash drought event or climatological characteristics of flash droughts using a single indicator (Ford and Labosier 2017; Koster et al. 2019; Christian et al. 2021). In these cases, even if other variables are assessed, the flash drought particulars of onset, duration, intensity, and spatial extent will be sensitive to the indicator used to characterize the event. The indicator-sensitivity is especially pernicious when assessing flash drought climatology on regional to global scales, as shown by Osman et al. (2021) and the EDDI comparison in this study (Fig. 6). In our case, the two different variations of EDDI-based flash drought indicators did not perform differently when compared with other indicators but did show large differences in regional frequency of flash drought. A climatological study of flash drought risk across CONUS using $\Delta EDDI_{7}$ would identify the Southeast and Southern Plains regions as "hotspots" with high flash drought frequency (as in Christian et al. 2019), whereas the EDDI-14 indicator would characterize these regions as having relatively low flash drought frequency. Such discrepancies make it challenging to study, communicate, plan, and prepare for flash drought.

Similarly, we find the climatological assessment and case study analysis results were sensitive to the method of flash drought identification, even when using the same indicator. In the 2012 and 2019 flash drought case studies, the EDDI-28 and SPEI-30 indicated flash drought in only a small fraction of the study areas. However, both indicators showed widespread drought conditions that expanded and intensified over a 4-6-week period. These results highlight issues that arise when 1) treating flash drought as a binary event (i.e., occurrence or no occurrence) in climatology- or risk-focused studies and 2) using a single method for identifying flash drought with one indicator. Based on the case study analyses, it is possible the previously published methods used for identifying flash drought with EDDI-28 and SPEI-30 may be underestimating the frequency of events that can practically be considered rapid onset droughts. Result sensitivity to flash drought identification methods is therefore also important to consider for flash drought research and assessment.

While multivariate, multi-indicator approaches to flash drought early warning, monitoring, and communication are preferred (Otkin et al. 2022), studies and early warning/ monitoring systems based on a single indicator can be useful to improving flash drought resilience if the biases, limitations, caveats of the indicator-and their implications for monitoring, early warning, and risk assessment-are well understood and communicated. Detailed assessments of flash drought indicators, datasets, and identification methods are particularly important in the context of evaluating projections and managing uncertainty of changes in flash drought frequency and severity in a warming climate because projected changes in flash drought risk greatly inform the development of adaptive management strategies (Ojima 2021). Therefore, future research on the historical or potential future hazard of flash drought should strive to either 1) use multiple indicators and variables to characterize the hazard; 2) test the sensitivity of the results

to different indicators, datasets, and methods; or 3) ensure comprehensive understanding and communication of the limitations and assumptions of the single indicator, dataset, or method used in the study.

In lieu of a single dataset recognized as the flash drought "truth" or benchmark by which to evaluate all indicators, assessments must incorporate multiple, diverse indicators, datasets, and methods for intercomparisons. Many studies have relied on the USDM to provide the benchmark by which to evaluate an indicator's flash drought monitoring or early warning prowess. However, we find the USDM consistently underperforms when compared with the other nine indicators, with the overall lowest TS values. These results are not necessarily surprising because the USDM was not designed to monitor flash drought, and it regularly includes lagging indicators such as streamflow and reservoir levels, which will respond more slowly to evolving drought than evaporative demand, precipitation, or soil moisture. We see these lagging characteristics of the USDM in the flash drought case study analyses as well (Figs. 7b and 8b). Therefore, the USDM is a good benchmark to ensure a flash drought indicator captures the drought part of the event, but it should not be used as the sole verification of any flash drought indicator. Our results suggest the USDM is not as ideally suited for flash drought monitoring across CONUS as other indicators evaluated.

b. Which indicator is "best"?

In this study, SPEI-14, SPI-14, and EDDI-14 had the highest overall TS values, attributable to the shorter window length and the indicators' ability to respond quicker to changing environmental conditions. These three indicators also consistently lead all others in depicting drought conditions in the 2012 and 2019 flash drought case studies. However, their relatively high biases suggest they may overestimate flash drought occurrence in most regions. For example, the SPEI-14 and EDDI-14 had bias ratio values of 1.97 and 1.84, respectively when compared with the soil moisture-based indicator (SMVI) in the Midwest region. The SPEI-14 and EDDI-14 indicate 1.8 and 1.5 flash drought events for every 1 event identified by SMVI in the Midwest, meaning 1 in every 3 flash droughts in EDDI-14 do not translate to soil moisture flash drought in the Midwest. Importantly, the flash drought frequency between indicators varied by region. For example, the same EDDI-14-SMVI comparison with a bias ratio of 1.84 in the Midwest had a bias ratio of only 1.02 in the Southeast, where the SMVI indicated a flash drought for every 1.1 flash droughts identified by EDDI-14. This suggests the rapidly increasing evaporative demand conditions captured by EDDI-14 more effectively translate to soil moisture depletion in the Southeast than in other regions. The regionality in the correspondence between indicators, such as EDDI-14 and SMVI, is a crucial consideration for researchers studying and recommending flash drought indicators for operational use across a region. In this case, our results would suggest that EDDI-14 may be a better fit for flash drought monitoring in the Southeast than in the Midwest.

Variations of EDDI, ESI, SPI, and SPEI indicators derived from 14-day periods consistently outperformed the

same indicators derived from 28- or 30-day periods based on TS. In the cases of EDDI and ESI, the 14-day derivations also provided a more consistent lead time of 2-6 days compared with other indicators, while the 28-day derivations tended to lag other indicators. These results were in agreement with our case study analyses as well, such that the 14-day EDDI, SPI, and SPEI consistently lead the 28- or 30-day derivations for both the 2012 and 2019 events. However, from a climatological assessment the 14-day SPI and SPEI did not consistently lead flash drought indications compared with 30-day SPI and SPEI. Additionally, while EDDI-14, SPEI-14, and SPI-14 had higher TS values than their 4-week or 30-day counterparts, they also exhibited much higher bias ratio scores. The lone exception was ESI-14, which had a higher TS score than its 4-week counterpart but maintained a bias ratio near 1. The performance differences between composite window length using the same dataset and methodology (i.e., SPI-14 vs SPI-30) demonstrate the sensitivity of results to averaging windows. The time over which indicators are compiled-for example, 2 weeks, 4 weeks, or 1 month-is an important component of the indicator and should be tailored to its use case.

Specifically, when taken across CONUS, the ESI-14 had a much lower bias ratio (1.03) relative to EDDI-14 (1.55), suggesting the inclusion of actual evapotranspiration with reference or potential evapotranspiration, as ESI does, may help reduce the number of evaporative demand-identified flash droughts that do not necessarily translate to soil moisture impacts. As with most of our findings, the difference of biases between ESI and EDDI varied regionally, with large differences in the Northeast, Midwest, northern plains, and Northwest, but much smaller differences in the Southeast, Southern Plains, and Southwest regions. These regional differences in EDDI-14 and ESI-14 performance may relate to how well changes in evaporative demand lead to changes in actual evapotranspiration, which is partly attributed to the coupling between land (soil and vegetation) and atmosphere (Dirmeyer et al. 2018). Therefore, indicators using only potential evapotranspiration may sufficiently forewarn subsequent soil moisture and vegetation response to emerging flash drought in regions with strong land-atmosphere coupling, but inclusion of actual evapotranspiration may be more ideal in energy-limited environments with weaker coupling between actual and potential evapotranspiration. Additionally, increasing evaporative demand as a function of anthropogenic global warming could affect the relationship between actual and potential evapotranspiration, with consequences for the representativeness of a solely demand-based indicator like EDDI (Tomas-Burguera et al. 2020). More research is needed to understand 1) the magnitude of evaporative demand changes with continued, hastened, or slowed future warming; and 2) how those changes could affect drought monitoring indicators like EDDI and ESI.

Our case study analyses revealed the indicators generally show agreement in rapidly intensifying drought conditions in both 2012 and 2019 flash drought events. However, there are important differences in the timing and extent of the various drought depictions. In both case studies, EDDI-14 showed increasing dryness 2–3 weeks earlier than other indicators, followed by SPI-14 and SPEI-14. The SPI-30, SPEI-30, and

USDM varied similarly through the evolution of both case study flash droughts and tended to lag other indicators by 2-3 weeks. The order of indicators showing drought expansion and flash drought occurrence generally follows the well documented physical processes underpinning flash drought across the United States (e.g., Otkin et al. 2018), with evaporative demand leading evapotranspiration leading soil moisture leading combined indicators like the U.S. Drought Monitor. A few indicators, including EDDI-28 and SPEI-30, did not show extensive flash drought in the 2012 and 2019 events, despite showing widespread drought conditions that intensified over time periods sufficiently short to practically be considered rapid. The lack of flash drought extent in these cases is therefore more likely attributable to fundamental limitations in the method of flash drought identification and less so to the indicator, itself.

c. Convergence toward diverse perspectives

Multivariable or multi-indicator approaches to flash drought research, early warning, monitoring, and assessment are preferred over the use of a single indicator (Otkin et al. 2022). As also argued in Osman et al. (2021), our results provide evidence that research should not try to converge to a single indicator from which to infer information about flash drought across all regions. Instead, the inconsistency and regional variability in our results suggest a variety of perspectives-including evaporative demand, precipitation, soil moisture, and vegetation response-can provide a more holistic, representative, and actionable account of flash drought for operational early warning and monitoring. Likewise, research should not depend on a single indicator from which to infer historical and potential future characteristics of flash drought. Instead, the current and future risk of flash drought and resultant adaptive management and response strategies should be based on a multivariable perspective on flash drought. Additionally, the noteworthy differences in climatological flash drought frequency between EDDI-14 and $\Delta EDDI_z$ are evidence for the use of multiple datasets and methods when using a single indicator, like EDDI. Hazard mitigation, planning, and risk assessment based on either of the maps in Fig. 6a would result in very different conclusions, especially for the Southern Plains and Southeast regions where $\Delta EDDI_7$ shows almost double the number of flash drought events of EDDI-14 over the 20-yr study period.

d. Caveats and limitations

Our study provides a novel intercomparison of flash drought indicators, many of which are used for operational flash drought monitoring. However, the results of this study come with multiple limitations and caveats, including the limited 20-yr study period. An ideal study period would span many decades to capture as many flash drought events as possible, but our study was limited by the availability of USDM and ESI indicators. Another limitation is the use of USDM for our analysis. Because the USDM is the only weekly updated product, its direct comparison with daily resolution indicators partly contributes to its relatively poor performance. With that said, we evaluate USDM from the perspective of operational flash drought monitoring, 1728

such that a product with a once-per-week update period will not be as responsive to rapid changes in precipitation, evaporative demand, or soil moisture as products that are updated daily. It is also worth noting that virtually all flash drought indicators identified 1–2 flash droughts per year per grid cell. The unusualness of flash drought celerity, by definition, would imply they are not necessarily frequent events. Therefore, while we cannot definitively conclude all indicators tested overestimated flash drought frequency, it is safe to say that only a fraction of the flash drought events identified by these indicators caused tangible and known drought impacts.

Flash drought identification methods were not identical across all drought indicators. We considered standardizing all indicators to the same dynamic range and using one method to identify flash drought to ensure the method of flash drought identification did not affect the results. However, flash drought studies, whether based on historical or projected climate information, often use different methods for identifying and characterizing flash drought, even if using the same variable, like precipitation or evaporative demand. Likewise, the myriad operational products that are part of the national or global drought monitoring infrastructure also have somewhat to very different methods for identifying and characterizing drought and flash drought. Because the ultimate goal of this study is to help improve operational flash drought monitoring through indicator evaluation and comparison, we eventually determined the best way forward was to evaluate each indicator as it is currently used or is proposed to be used for operational monitoring.

Last, our intercomparison was done without considering flash drought impacts. Ideally, indicators should be evaluated with documented flash drought impacts, such as water restrictions, ecological damage, or below trend crop yields. Drought impacts have always been a challenge to integrate into research and assessment. However, the advent of effective drought impact monitoring and reporting infrastructure portends an improved capability to use impacts directly when evaluating drought monitoring (Stahl et al. 2016; Otkin et al. 2018). All future research developing new or improved flash drought indicators should—as much as possible—integrate documented flash drought impacts in their assessments. Last, the regional averages of indicator comparisons may obscure other important differences in indicator performance within a given region.

6. Conclusions

We evaluate flash drought monitoring capability across CONUS through an intercomparison of nine flash drought indicators, most of which are currently used for operational monitoring. The indicators represent diverse perspectives and processes that are known to drive flash drought, including evapotranspiration and evaporative demand (ESI and EDDI), precipitation (SPEI and SPI), and soil moisture (SMVI). We also include the USDM, which is frequently used as a benchmark by which to evaluate flash drought indicators.

We find the precipitation and evaporative demand indicators derived over 14-day periods tended to have the highest correspondence overall to the other indicators and had the longest lead time, but also the highest flash drought frequency. The lone exception with a bias ratio closer to an ideal value of 1 was ESI-14, which tended to produce fewer flash droughts than similar indicators that do not incorporate actual evapotranspiration (e.g., EDDI-14 and SPEI-14). The 14-day SPI and SPEI indicators had consistently higher correspondence (TS) and higher bias ratio than their 30-day counterparts but did not provide consistently longer lead times for early warning. The USDM had the overall lowest correspondence with other indicators, which is partly attributable to the weekly update frequency of the USDM, relative to the daily update rates of the other indicators. The USDM also incorporates inherently lagging indicators of drought—such as streamflow and reservoir levels—that will not necessarily reflect flash drought as quickly as other indicators.

Importantly, we find considerable differences in indicator performance between CONUS regions. EDDI-14 had very high flash drought frequency in most regions, but less so in the Southeast and Southwest. We evaluated the sensitivity of the results to the definition used for identifying flash drought using an alternative method based on 14-day EDDI. The results showed large differences in the climatological frequency of flash drought between EDDI-14 and the alternative EDDI formation (Δ EDDI_z). The results suggest that 1) flash drought characteristics and climatological frequency (risk) are very sensitive to the definition of flash drought based on a single indicator (like EDDI) and 2) the sensitivity varies regionally.

Overall, we find no single flash drought indicator consistently outperforms across CONUS or any of the seven regions. Likewise, the regional and definition-specific variability in performance supports the argument for a multivariable and multi-indicator approach for flash drought early warning and monitoring advocated by Otkin et al. (2022). Flash drought research, especially evaluation of historical and potential future changes in flash drought climatological risk and characteristics, should seek multivariable frameworks for analysis, instead of using a single indicator from which to infer all flash drought information. In doing so, researchers can properly understand and communicate the complexity of flash drought and its impacts, and better guide effective adaptive management and response strategies.

Acknowledgments. This work was supported by the National Integrated Drought Information System (NIDIS Award SUBAWD002197) and National Oceanic and Atmospheric Administration (NOAA Award NA22OAR4310135). Thanks are given to Mike Hobbins for his suggestions and help with EDDI.

Data availability statement. EDDI, ESI, SMVI, SPI, SPEI, and USDM data used to identify flash droughts across the contiguous United States are made available online (https://drive.google.com/drive/folders/17f6l_6Q_8ZJO_4vouOx05N6 GCQNaf2y1?usp=sharing).

REFERENCES

Abatzoglou, J. T., 2013: Development of gridded surface meteorological data for ecological applications and modeling. *Int. J. Climatol.*, 33, 121–131, https://doi.org/10.1002/joc.3413.

- Allen, R. G., A. J. Clemmens, C. M. Burt, K. Solomon, and T. O'Halloran, 2005: Prediction accuracy for projectwide evapotranspiration using crop coefficients and reference evapotranspiration. J. Irrig. Drain. Eng., 131, https://doi.org/10.1061/(ASCE) 0733-9437(2005)131:1(24).
- Anderson, M. C., J. M. Norman, G. R. Diak, W. P. Kustas, and J. R. Mecikalski, 1997: A two-source time-integrated model for estimating surface fluxes using thermal infrared remote sensing. *Remote Sens. Environ.*, **60**, 195–216, https://doi.org/ 10.1016/S0034-4257(96)00215-5.
- —, —, J. R. Mecikalski, J. A. Otkin, and W. P. Kustas, 2007: A climatological study of evapotranspiration and moisture stress across the continental United States based on thermal remote sensing: 1. Model formulation. J. Geophys. Res., 112, 2006JD007506, https://doi.org/10.1029/2006JD007506.
- —, C. Hain, J. Otkin, X. Zhan, K. Mo, M. Svoboda, B. Wardlow, and A. Pimstein, 2013: An intercomparison of drought indicators based on thermal remote sensing and NLDAS-2 simulations with U.S. Drought Monitor classifications. *J. Hydrometeor.*, 14, 1035–1056, https://doi.org/10.1175/JHM-D-12-0140.1.
- Bachmair, S., C. Svensson, J. Hannaford, L. J. Barker, and K. Stahl, 2016: A quantitative analysis to objectively appraise drought indicators and model drought impacts. *Hydrol. Earth Syst. Sci.*, **20**, 2589–2609, https://doi.org/10.5194/hess-20-2589-2016.
- Basara, J. B., J. I. Christian, R. A. Wakefield, J. A. Otkin, E. H. Hunt, and D. P. Brown, 2019: The evolution, propagation, and spread of flash drought in the central United States during 2012. *Environ. Res. Lett.*, 14, 084025, https://doi.org/10. 1088/1748-9326/ab2cc0.
- Boyer, J. S., and Coauthors, 2013: The U.S. drought of 2012 in perspective: A call to action. *Global Food Secur.*, 2, 139–143, https://doi.org/10.1016/j.gfs.2013.08.002.
- Chen, L. G., J. Gottschalck, A. Hartman, D. Miskus, R. Tinker, and A. Artusa, 2019: Flash drought characteristics based on U.S. Drought Monitor. *Atmosphere*, **10**, 498, https://doi.org/ 10.3390/atmos10090498.
- Christian, J. I., J. B. Basara, J. A. Otkin, E. D. Hunt, R. A. Wakefield, P. X. Flanagan, and X. Xiao, 2019: A methodology for flash drought identification: Application of flash drought frequency across the United States. J. Hydrometeor., 20, 833– 846, https://doi.org/10.1175/JHM-D-18-0198.1.
- —, —, E. D. Hunt, J. A. Otkin, and X. Xiao, 2020: Flash drought development and cascading impacts associated with the 2010 Russian heatwave. *Environ. Res. Lett.*, **15**, 094078, https://doi.org/10.1088/1748-9326/ab9faf.
- —, —, —, , J. C. Furtado, V. Mishra, X. Xiao, and R. M. Randall, 2021: Global distribution, trends, and drivers of flash drought occurrence. *Nat. Commun.*, **12**, 6330, https:// doi.org/10.1038/s41467-021-26692-z.
- Di Liberto, T., 2019: Flash drought engulfs the U.S. Southeast in September 2019. NOAA Climate.gov News & Features, https://www.climate.gov/news-features/event-tracker/flash-droughtengulfs-us-southeast-september-2019.
- Dirmeyer, P. A., and Coauthors, 2018: Verification of land-atmosphere coupling in forecast models, reanalyses, and land surface models using flux site observations. J. Hydrometeor., 19, 375–392, https://doi.org/10.1175/JHM-D-17-0152.1.
- Ebert, E. E., 2008: Fuzzy verification of high-resolution gridded forecasts: A review and proposed framework. *Meteor. Appl.*, 15, 51–64, https://doi.org/10.1002/met.25.

- Ford, T. W., and C. F. Labosier, 2017: Meteorological conditions associated with the onset of flash drought in the eastern United States. *Agric. For. Meteor.*, 247, 414–423, https://doi. org/10.1016/j.agrformet.2017.08.031.
- —, and S. M. Quiring, 2019: Comparison of contemporary in situ, model, and satellite remote sensing soil moisture with a focus on drought monitoring. *Water Resour. Res.*, **55**, 1565– 1582, https://doi.org/10.1029/2018WR024039.
- —, D. B. McRoberts, S. M. Quiring, and R. E. Hall, 2015: On the utility of in situ soil moisture observations for flash drought early warning in Oklahoma, USA. *Geophys. Res. Lett.*, 42, 9790–9798, https://doi.org/10.1002/2015GL066600.
- Fu, K., and K. Wang, 2022: Quantifying flash droughts over China from 1980 to 2017. J. Geophys. Res. Atmos., 127, e2022JD037152, https://doi.org/10.1029/2022JD037152.
- Fuchs, B., and Coauthors, 2012: From too much to too little: How the central U.S. drought of 2012 evolved out of one of the most devastating floods on record in 2011. Publications of the National Drought Mitigation Center 5, 99 pp., https:// digitalcommons.unl.edu/ndmcpub/5.
- Gutzmer, D., 2018: Christmas tree shortages across the country attributed partly to drought. National Drought Mitigation Center News, https://drought.unl.edu/Publications/News.aspx?id=332.
- Hobbins, M. T., A. Wood, D. J. McEvoy, J. L. Huntington, C. Morton, M. Anderson, and C. Hain, 2016: The evaporative demand drought index. Part I: Linking drought evolution to variations in evaporative demand. J. Hydrometeor., 17, 1745–1761, https://doi.org/10.1175/JHM-D-15-0121.1.
- Hoell, A., and Coauthors, 2020: Lessons learned from the 2017 flash drought across the U.S. northern Great Plains and Canadian Prairies. *Bull. Amer. Meteor. Soc.*, **101**, E2171–E2185, https://doi.org/10.1175/BAMS-D-19-0272.1.
- Hoffmann, D., A. J. E. Gallant, and M. Hobbins, 2021: Flash drought in CMIP5 models. J. Hydrometeor., 22, 1439–1454, https://doi.org/10.1175/JHM-D-20-0262.1.
- Hunt, E., and Coauthors, 2021: Agricultural and food security impacts from the 2010 Russia flash drought. Wea. Climate Extremes, 34, 100383, https://doi.org/10.1016/j.wace.2021.100383.
- Kam, J., S. Kim, and J. K. Roundy, 2021: Did a skillful prediction of near-surface temperatures help or hinder forecasting of the 2012 US drought? *Environ. Res. Lett.*, 16, 034044, https:// doi.org/10.1088/1748-9326/abe1f6.
- Koster, R. D., S. D. Schubert, H. Wang, S. P. Mahanama, and A. M. DeAngelis, 2019: Flash drought as captured by reanalysis data: Disentangling the contributions of precipitation deficit and excess evapotranspiration. J. Hydrometeor., 20, 1241–1258, https://doi.org/10.1175/JHM-D-18-0242.1.
- Lisonbee, J., Woloszyn, M., and M. Skumanich, 2021: Making sense of flash drought: Definitions, indicators, and where we go from here. J. Appl. Serv. Climatol., 2021, https://doi.org/10. 46275/JOASC.2021.02.001.
- McKee, T. B., N. J. Doesken, and J. Kleist, 1993: The relationship of drought frequency and duration to time scales. *Eighth Conf. on Applied Climatology*, Anaheim, CA, Amer. Meteor. Soc., 179–184.
- Mozny, M., M. Trnka, Z. Zalud, P. Hlavinka, J. Nekovar, V. Potop, and M. Virag, 2012: Use of a soil moisture network for drought monitoring in the Czech Republic. *Theor. Appl. Climatol.*, **107**, 99–111, https://doi.org/10.1007/s00704-011-0460-6.
- Nguyen, H., M. C. Wheeler, J. A. Otkin, T. Cowan, A. Frost, and R. Stone, 2019: Using the evaporative stress index to monitor flash drought in Australia. *Environ. Res. Lett.*, **14**, 064016, https://doi.org/10.1088/1748-9326/ab2103.

--, ---, T. Nguyen-Huy, and T. Cowan, 2023: Climatology and composite evolution of flash drought over Australia and its vegetation impacts. J. Hydrometeor., 24, 1087–1101, https://doi.org/10.1175/JHM-D-22-0033.1.

- Noguera, I., F. Dominguez-Castro, and S. M. Vicente-Serrano, 2020: Characteristics and trends of flash droughts in Spain, 1961–2018. Ann. N. Y. Acad. Sci., 1472, 155–172, https://doi. org/10.1111/nyas.14365.
- —, S. M. Vicente-Serrano, and F. Dominguez-Castro, 2022: The rise of atmospheric evaporative demand is increasing flash droughts in Spain during the warm season. *Geophys. Res. Lett.*, **49**, e2021GL097703, https://doi.org/10.1029/2021GL097703.
- O, S., and S. K. Park, 2023: Flash drought drives rapid vegetation stress in arid regions in Europe. *Environ. Res. Lett.*, 18, 014028, https://doi.org/10.1088/1748-9326/acae3a.
- Ojima, D. S., 2021: Climate resilient management in response to flash droughts in the US northern Great Plains. *Curr. Opin. Environ. Sustainability*, **48**, 125–131, https://doi.org/10.1016/j. cosust.2021.01.002.
- Osman, M., B. F. Zaitchik, H. S. Badr, J. I. Christian, T. Tadesse, J. A. Otkin, and M. C. Anderson, 2021: Flash drought onset over the contiguous United States: Sensitivity of inventories and trends to quantitative definitions. *Hydrol. Earth Syst. Sci.*, 25, 565–581, https://doi.org/10.5194/hess-25-565-2021.
- —, and Coauthors, 2022: Diagnostic classification of flash drought events reveals distinct classes of forcings and impacts. J. Hydrometeor., 23, 275–289, https://doi.org/10.1175/ JHM-D-21-0134.1.
- Otkin, J. A., M. C. Anderson, C. Hain, I. E. Mladenova, J. B. Basara, and M. Svoboda, 2013: Examining rapid onset drought development using the thermal infrared-based evaporative stress index. J. Hydrometeor., 14, 1057–1074, https://doi.org/ 10.1175/JHM-D-12-0144.1.
- —, and Coauthors, 2016: Assessing the evolution of soil moisture and vegetation conditions during the 2012 United States flash drought. *Agric. For. Meteor.*, **218–219**, 230–242, https:// doi.org/10.1016/j.agrformet.2015.12.065.
- —, M. Svoboda, E. D. Hunt, T. W. Ford, M. C. Anderson, C. Hain, and J. B. Basara, 2018: Flash droughts: A review and assessment of the challenges imposed by rapid-onset droughts in the United States. *Bull. Amer. Meteor. Soc.*, **99**, 911–919, https://doi.org/10.1175/BAMS-D-17-0149.1.
- —, Y. Zhong, E. D. Hunt, J. Basara, M. Svoboda, M. C. Anderson, and C. Hain, 2019: Assessing the evolution of soil moisture and vegetation conditions during a flash drought-flash recovery sequence over the south-central United States. J. Hydrometeor., 20, 549–562, https://doi.org/10.1175/JHM-D-18-0171.1.
- —, and Coauthors, 2021: Development of a flash drought intensity index. *Atmosphere*, **12**, 741, https://doi.org/10.3390/ atmos12060741.

- —, and Coauthors, 2022: Getting ahead of flash drought: From early warning to early action. *Bull. Amer. Meteor. Soc.*, 103, E2188–E2202, https://doi.org/10.1175/BAMS-D-21-0288.1.
- Parker, T., A. Gallant, M. Hobbins, and D. Hoffmann, 2021: Flash drought in Australia and its relationship to evaporative demand. *Environ. Res. Lett.*, 16, 064033, https://doi.org/10. 1088/1748-9326/abfe2c.
- Pendergrass, A. G., and Coauthors, 2020: Flash droughts present a new challenge for subseasonal-to-seasonal prediction. *Nat. Climate Change*, **10**, 191–199, https://doi.org/10.1038/s41558-020-0709-0.
- Rippey, B. R., 2015: The U.S. drought of 2012. *Wea. Climate Extremes*, **10**, 57–64, https://doi.org/10.1016/j.wace.2015.10.004.
- Schwartz, C. S., and R. A. Sobash, 2017: Generating probabilistic forecasts from convection-allowing ensembles using neighborhood approaches: A review and recommendations. *Mon. Wea. Rev.*, **145**, 3397–3418, https://doi.org/10.1175/MWR-D-16-0400.1.
- Sobash, R. A., J. S. Kain, D. R. Bright, A. R. Dean, M. C. Coniglio, and S. J. Weiss, 2011: Probabilistic forecast guidance for severe thunderstorms based on the identification of extreme phenomena in convection-allowing model forecasts. *Wea. Forecasting*, 26, 714–728, https://doi.org/10.1175/WAF-D-10-05046.1.
- Stahl, K., and Coauthors, 2016: Impacts of European drought events: Insights from an international database of text-based reports. *Nat. Hazards Earth Syst. Sci.*, 16, 801–819, https://doi. org/10.5194/nhess-16-801-2016.
- Svoboda, M., and Coauthors, 2002: The Drought Monitor. Bull. Amer. Meteor. Soc., 83, 1181–1190, https://doi.org/10.1175/ 1520-0477-83.8.1181.
- Tomas-Burguera, M., S. M. Vicente-Serrano, D. Pena-Angulo, F. Dominguez-Castro, I. Noguera, and A. El Kenawy, 2020: Global characterization of the varying responses of the standardized precipitation evapotranspiration index to atmospheric evaporative demand. J. Geophys. Res. Atmos., 125, e2020JD033017, https://doi.org/10.1029/2020JD033017.
- Vicente-Serrano, S. M., S. Begueria, and J. I. López-Moreno, 2010: A multiscalar drought index sensitive to global warming: The standardized precipitation evapotranspiration index. *J. Climate*, 23, 1696–1718, https://doi.org/10.1175/2009JCLI2909.1.
- Xia, Y., and Coauthors, 2012: Continental-scale water and energy flux analysis and validation for the North American land data assimilation system project phase 2 (NLDAS-2): 1. Intercomparison and application of model products. J. Geophys. Res., 117, D03109, https://doi.org/10.1029/2011JD016048.
- —, M. B. Ek, Y. Wu, T. Ford, and S. M. Quiring, 2015: Comparison of NLDAS-2 simulated and NASMD observed daily soil moisture. Part I: Comparison and analysis. J. Hydrometeor., 16, 1962–1980, https://doi.org/10.1175/JHM-D-14-0096.1.